Author's response to reviews

Title: Human Subjects Protection Issues in QUERI Implementation Research

Authors:

Edmund Chaney (chaney@u.washington.edu)
Laura G. Rabuck (Laura.Rabuck@va.gov)
Jane Uman (Jane.Uman@va.gov)
Debbie Mittman (Debbie.Mittman@va.gov)
Carol Simons (Carol.Simons@va.gov)
Barbara Simon (Barbara.Simon@va.gov)
Mona Ritchie (RitchieMonaJ@uams.edu)
Marisue Cody (Marisue.Cody@va.gov)
Lisa V Rubenstein (lisar@rand.org)

Version: 4 Date: 26 September 2007

Author's response to reviews: see over
September 25, 2007

Martin Eccles
Ian Graham
Series Editors
Implementation Science

Re: MS ID 2717689201122955
Title: Human Subjects Protection Issues in QUERI Implementation Research
Journal: Implementation Science
Authors: Edmund Chaney, Laura G. Rabuck, Jane Uman, Debbie Mittman, Carol Simons, Barbara Simon, Mona Ritchie, Marisue Cody and Lisa V. Rubenstein

Dear Dr. Eccles and Graham:

We have responded to all of the copy editing suggestions dated September 7, 2007 and either accepted them or retained the original text where the meaning would have been changed. Thanks again for your feedback on our manuscript. Please contact us if there are any further revisions.

Sincerely,

Edmund Chaney, Ph.D.
Corresponding Author